Author's response to reviews

Title: Study of the therapeutic effects of an advanced hippotherapy simulator in children with cerebral palsy: a randomised controlled trial.

Authors:

Pablo Herrero (pherrero@usj.es)
Angel Asensio (aasensio@unizar.es)
Elena García (elenaga@aidimo.org)
Álvaro Marco (amarco@unizar.es)
Bárbara Olivan (bolivan@usj.es)
Alejandro Ibarz (aibarz@unizar.es)
Eva María Gómez-Trullén (evagomez@unizar.es)
Roberto Casas (rcasas@unizar.es)

Version: 2 Date: 12 February 2010

Author's response to reviews: see over
COVER LETTER

According to the reviewer’s guidelines we have carried out the following changes in the article:

1. We have added a discussion part in the abstract, which remains shorter than 350 words.
2. We have eliminated abbreviations in the abstract.
3. We have reviewed the methods section of the manuscript including details of sample size calculation.
4. We have added a statement in the Methods section of the manuscript, including the name of the body which gave Ethics approval.
5. We have also added a statement explaining that informed consent would be requested from children’s parents or tutors.